Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: This case report is a good piece of work presenting an interesting clinical case and method of its management. The article will be a valuable source of knowledge for all clinicians who meet similar problems. Theoretic introduction is prepared carefully and systematically and clinical diagnostic and therapeutic part is described in detail, valuable also from a didactic point of view. For these reasons, I recommend this article for publication in World Journal of Clinical Cases.

**ANSWER:** We appreciate your positive comment very much. The language has been further polished carefully.

Reviewer #2:
Scientific Quality: Grade A (Excellent)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: This manuscripts was well organized. What I suggest to authors is to identify the chemical compound which the Honeysuckle contains.

**ANSWER:** Thanks for your positive comment. The main chemical compound of Honeysuckle are phenolic acids, including chlorogenic acid, neochlorogenic acid and cryptochlorogenic acid, which have been supplemented in the manuscript.

**References:** Fang Wang, Cunyu Li,Yunfeng Zheng, Yu Li,Guoping Peng. Study on the anaphylactoid of three phenolic acids in Honeysuckle. J Ethnopharmacol 2015;170:1-7[PMID: 25978951 DOI: 10.1016/j.jep.2015.05.011]